\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}



\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{510.67, 512.545} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{03C62, 06F15} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem{remark}{Remark}
\newtheorem{example}{Example}


\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue,
citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
     {\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 21, No. ?, pp. 144--145 (2024) \hspace{\fill}{\rm\small УДК \udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
        {\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2024.21.???}}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \ \mscs }
  }%\vbox
}




\begin{document}
\begin{thebibliography}{99}

\bibitem{Fuchs} L.~Fuchs, \href{https://books.google.ru/books/about/Partially_Ordered_Algebraic_Systems.html?id=V_k79sVPcqYC&redir_esc=y}{\it
Partially Ordered Algebraic Systems}, Dover books on mathematics,
International series of monographs in pure and applied
mathematics, Courier Corporation, 2011. 229~p.

\bibitem{KoKo} A.I.~Kokorin, V.M. Kopytov \href{https://books.google.ru/books/about/Fully_Ordered_Groups.html?id=7t2AAAAAIAAJ&redir_esc=y} {\it Fully ordered groups},
New Work, J.~Wiley, 1974. 147~p.

\bibitem{KoMe} V.M. Kopytov, N.Ya. Medvedev \href{https://doi.org/10.1007/978-94-015-8304-6} {\it The theory of lattice-ordered groups},
Dordrecht, Springer, 1994. 400~p.

\bibitem{Shepperd1} J.A.H. Shepperd, \href{https://doi.org/10.1112/jlms/s1-31.2.240} {\it Separation and the definition of betweenness and separation
groups}, Journal of the London Mathematical Society,
\textbf{s1-31}:2 (1956), 240--248.

\bibitem{Shepperd2} J.A.H. Shepperd, \href{https://doi.org/10.1112/jlms/s1-32.3.277} {\it Betweenness groups}, Journal of the London Mathematical Society, \textbf{32} (1957), 277--285.

\bibitem{Shepperd3} J.A.H. Shepperd, \href{https://doi.org/10.1112/plms/s3-7.1.518} {\it Separation groups}, Proceedings of the London Mathematical Society, \textbf{s3-7}:1 (1957), 518--548.

\bibitem{Gilder} J. Gilder, \href{https://doi.org/10.1017/s0305004100038585} {\it Betweenness and order in semigroups}.
Mathematical Proceedings of the Cambridge Philosophical Society,
\textbf{61}:1 (1965), 13--28.

\bibitem{aafot} S.V.~Sudoplatov, \href{https://doi.org/10.33048/semi.2022.19.075} {\it Arities and aritizabilities of first-order
theories}, Siberian Electronic Mathematical Reports, \textbf{19}:2
(2022), 889--901.

\bibitem{dso} B.Sh.~Kulpeshov, S.V.~Sudoplatov, \href{https://doi.org/10.33048/semi.2023.20.034} {\it Spherical orders, properties and countable spectra of their
theories}, Siberian Electronic Mathematical Reports, \textbf{20}:2
(2023), 588--599.

\bibitem{mcso} S.V.~Sudoplatov, \href{https://doi.org/10.33048/semi.2023.20.035} {\it Minimality conditions, topologies and ranks for spherically ordered theories},
Siberian Electronic Mathematical Reports, \textbf{20}:2 (2023),
600--615.

\bibitem{Mul} K. Muliarchyk, \href{https://doi.org/10.48550/arXiv.2403.14779} {\it Free products of bi-orderable
groups}, arXiv:2403.14779 [math.GR], 2024. 15~p.


\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}


\thispagestyle{empty}

\title[Spherically orderable groups]{Spherically orderable groups}

\author[S.V. SUDOPLATOV]{{\bf S.V. Sudoplatov}\protect\orcidicon{0000-0002-3268-9389}}


\address{Sergey Vladimirovich Sudoplatov  % Контактные данные всех авторов и место работы указываются только на английском языке
\newline\hphantom{iii} Sobolev Institute of Mathematics,
\newline\hphantom{iii} Academician Koptyug avenue, 4
\newline\hphantom{iii} 630090, Novosibirsk, Russia}%
\email{\textcolor{blue}{sudoplat@math.nsc.ru}}%


\thanks{\sc Sudoplatov, S.V.,
Spherically orderable groups}
\thanks{\copyright \ 2024 Sudoplatov S.V.}
\thanks{\rm The work
was carried out in the framework of the State Contract of the
Sobolev Institute of Mathematics, Project No.~FWNF-2022-0012.}
\thanks{\it Received January, 1, 2024, Published December, 31, 2024}%


\semrtop \vspace{1cm} \maketitle {\small

\vspace{-12pt} \centerline{{\it Communicated by} {\sc P.P. Petrov}
%({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} We introduce and study the class of
spherically ordered groups. The notions of spherically orderable
groups and their spec\-t\-ra of spherical orderability are
defined. Values of these spectra are found for a series of natural
groups.

\medskip
\noindent{\bf Keywords:} spherical order, group, spectrum of
spherical orderability.
 \end{quote}
}


\bigskip
\section{Introduction}

Well known linearly ordered groups and circularly ordered groups
are both deeply investigated and described \cite{Fuchs, KoKo} and
admit various generalizations and modifications for partial, left
and right  orderings \cite{Fuchs, KoMe}, betweenness and
separation groups \cite{Shepperd1, Shepperd2, Shepperd3} and
semigroups \cite{Gilder}.

We continue to study $n$-spherical orders introducing and
investigating $n$-sphe\-ri\-cal\-ly ordered groups which
generalize linearly and cyclically ordered groups, with $n=2$ and
$n=3$, respectively. The notions of spherically ordered groups and
their spectra of spherical orderability are introduced. Values of
these spectra are found for a series of natural groups.

\section{Spherically orderable groups and their spherical spectra}

Recall the following generalization of linear and circular orders.

\medskip
{\bf Definition} \cite{aafot, dso, mcso}. An $n$-ary relation
$K_n$ is called {\em $n$-ball}, or {\em $n$-spherical}, or {\em
$n$-circular} order relation, for $n\geq 2$, if it satisfies the
following conditions:

(nso1)\footnote{Here we admit cyclic permutations iff $n$ is odd. In view of this circumstance the item (nso1) is corrected with respect to the previous version.} for any even permutation $\sigma$ on $\{1,2,\ldots,n\}$, $$\forall x_1,\ldots,x_n \left(K_n(x_1,x_2,\ldots,x_n)\to K_n\left(x_{\sigma(1)},x_{\sigma(2)}\ldots,x_{\sigma(n)}\right)\right);$$\\

(nso2) $\forall x_1,\ldots,x_n
\biggm((K_n(x_1,\ldots,x_i,\ldots,x_j,\ldots,x_n)\land$
$$\land K_n(x_1,\ldots,x_j,\ldots,x_i,\ldots,x_n))\leftrightarrow\bigvee\limits_{1\leq k<l\leq n} x_k= x_l\biggm)$$ for any $1\leq i<j\leq n$;\\

(nso3) $\forall x_1,\ldots,x_n\Biggm(K_n(x_1,\ldots,x_n)\to$
$$\to\forall t\left(\bigvee\limits_{i=1}^nK_n(x_1,\ldots,x_{i-1},t,x_{i+1},\ldots,x_n)\right)\Biggm);$$\\

(nso4) $\forall x_1,\ldots,x_n
(K_n(x_1,\ldots,x_i,\ldots,x_j,\ldots,x_n)\lor$
$$\lor K_n(x_1,\ldots,x_j,\ldots,x_i,\ldots,x_n)),\,\, 1\leq i<j\leq
 n.$$

A structure $\mathcal{M}$ provided with a $n$-spherical order is
called {\em $n$-spherically ordered}.

\medskip
The axioms above produce all possible linear orders $K_2$ and
circular orders $K_3$. Here (nso2) gives the reflexivity: $\forall
x K_2(x,x)$, and the antisymmetry: $$\forall
x_1,x_2(K_2(x_1,x_2)\wedge K_2(x_2,x_1)\to x_1\approx x_2),$$
(nso2) and (nso3) give the transitivity: $$\forall
x_1,x_2,x_3(K_2(x_1,x_2)\wedge K_2(x_2,x_3)\to K_2(x_1,x_3)),$$
and the axiom (nso4) gives the linearity: $\forall
x_1,x_2(K_2(x_1,x_2)\vee K_2(x_2,x_1)).$ For the transitivity it
suffices to take pairwise distinct elements $a,b,c$ with $(a,b)\in
K_2$ and $(b,c)\in K_2$. By (nso3) we have $(a,c)\in K_2$ or
$(c,b)\in K_2$, and $(b,a)\in K_2$ or $(a,c)\in K_2$. But the
cases $(c,b)\in K_2$ and $(b,a)\in K_2$ are impossible in view of
(nso2), implying the required $(a,c)\in K_2$.

Similarly to linear orders any $n$-spherical order $K_n$ on a set
$Z$ has the dual one consisting of all $n$-tuples in $Z^n\setminus
K_n$ united with the set of all $n$-tuples in $Z^n$ with some
repeated coordinates. We denote this dual order by
$\overline{K_n}$.

\medskip
{\bf Definition.} A group $G$ is called {\em $n$-spherically
ordered}, or {\em $n$-$s$-ordered}, if $G$ is provided with a
$n$-spherical order $K_n$ such that for any $(x_1,\ldots,x_n)\in
K_n$ and any $y\in G$ the tuples $(x_1y,\ldots,x_ny)$ and
$(yx_1,\ldots,yx_n)$ belong to $K_n$.

A group $G$ is called {\em $n$-spherically orderable}, or {\em
$n$-$s$-orderable}, if $G$ has a $n$-spherically ordered
expansion. A group $G$ is called {\em spherically orderable} if it
is $n$-spherically orderable for some $n$.

For a group $G$ we define its {\em spectrum ${\rm Sp}_{\rm so}$ of
spherical orderability}, or {\em spherical spectrum}, as follows:
$$
{\rm Sp}_{\rm so}(G)=\{n\in\omega\setminus\{0,1\}\mid G\mbox{ is
}n\mbox{-spherically orderable}\}.
$$

A group $G$ is called {\em totally spherically orderable}, or {\em
totally $s$-orderable}, if $G$ has maximal spectrum of spherical
orderability, i.e. ${\rm Sp}_{\rm so}(G)=\omega\setminus\{0,1\}$.

A group $G$ is called {\em almost totally spherically orderable},
or {\em almost totally $s$-orderable}, if  ${\rm Sp}_{\rm so}(G)$
is a cofinite subset of $\omega$.

A group $G$ is ({\em almost}) {\em not $s$-orderable in any way}
if ${\rm Sp}_{\rm so}(G)$ is empty (respectively, finite).

\medskip
The notions above for the spherical orderability and its spectra
can be naturally spread for an arbitrary structure. Besides, the
spherical orderability admits similar variations of orderability
as for linear (bi-)orderability such as left-orderability and
right-orderability \cite{KoMe, Mul}.

A natural {\bf problem} arises on description of spherical spectra
for groups and related structures.

\medskip
By the definition a group $G$ is linearly ordered iff $G$ is
$2$-spherically ordered, and $G$ is cyclically ordered iff $G$ is
$3$-spherically ordered. Here $2\in{\rm Sp}_{\rm so}(G)$ and
$3\in{\rm Sp}_{\rm so}(G)$, respectively.

Again by the definition a group $G$ is spherically orderable iff
${\rm Sp}_{\rm so}(G)\ne\emptyset$.

\begin{remark}\label{rem_spher0}\rm
If a group $G$ is $n$-spherically orderable then each subgroup of
$G$ is $n$-spherically orderable, too, since any restriction
$\mathcal{M}$ of a $n$-spherically ordered structure
$\mathcal{N}$, with a $n$-spherical order $K_n$, is again
$n$-spherically ordered, with the $n$-spherical order $K_n\cap
M^n$.
\end{remark}

In view of Remark \ref{rem_spher0} we have the following {\em
Monotonicity property} for the spectrum of spherical orderability:

\begin{proposition}\label{pr_sp_mon}
For any groups $G_1, G_2$ if $G_1\leqslant G_2$ then ${\rm
Sp}_{\rm so}(G_1)\supseteq{\rm Sp}_{\rm so}(G_2)$.
\end{proposition}

Proposition \ref{pr_sp_mon} immediately implies:

\begin{corollary}\label{cor_tot}
If $G$ is an {\rm (}almost{\rm )} totally $s$-orderable group then
any its subgroup is also {\rm (}almost{\rm )} totally
$s$-orderable.
\end{corollary}

\begin{corollary}\label{cor_nordaw}
If $G$ is {\rm (}almost{\rm )} not $s$-orderable in any way then
any its supergroup is also {\rm (}almost{\rm )} not $s$-orderable
in any way.
\end{corollary}

Proposition \ref{pr_sp_mon} and Corollary \ref{cor_tot} can be
also deduced from the following criterion of $n$-spherical
orderability:

\begin{theorem}\label{th_sp_cr}
A group $G$ is $n$-spherically ordered by a $n$-spherical order
$K_n$ iff for any $n$-tuple $(a_1,\ldots,a_n)\in K_n$ with
pairwise distinct coordinates and for any $b\in G$ the tuples
$(a_1b,\ldots,a_nb)$ and $(ba_1,\ldots,ba_n)$ are even
permutations, i.e. are not odd permutations of some tuples in
$K_n$.
\end{theorem}

Proof. By the definition of $n$-spherical order $K_n$ it consists
of all $n$-tuples with some repeated coordinates and of all even
permutations of given $n$-tuples in $K_n$ such that for any
$n$-tuple, in the universe, with pairwise distinct coordinate
either this tuple or all its odd permutations belong to $K_n$. Now
if $G$ is $n$-spherically ordered by $K_n$ it is forbidden to
include into $K_n$ its odd permutations of forms
$(a_1b,\ldots,a_nb)$ and $(ba_1,\ldots,ba_n)$, and conversely
permitted to include its even permutations of forms
$(a_1b,\ldots,a_nb)$ and $(ba_1,\ldots,ba_n)$. \endproof

\begin{remark}\label{rem_sp_perm}\rm
Theorem \ref{th_sp_cr} shows that the only obstacle for a
$n$-spherical order $K_n$ on a group $G$ to produce this group to
be $n$-spherical ordered is the possibility for the
multiplications $(a_1b,\ldots,a_nb)$ and $(ba_1,\ldots,ba_n)$ to
produce an odd permutation for a tuple $(a'_1,\ldots,a'_n)$ in
$K_n$ with pairwise distinct coordinates. In particular, elements
of stabilizers for $n$-element sets $\{a_1,\ldots,$
$a_n\}\subseteq G$ can not produce odd permutations of tuples
$(a_1,\ldots,a_n)\in K_n$. Here $b\ne e$ and therefore the maps
$a_i\mapsto a_ib$ and $a_i\mapsto ba_i$ do not have fixed
elements.

In general, $K_n$ is uniquely defined by its subrelation $K^0_n$
consisting of all $n$-tuples in $K_n$ with pairwise distinct
coordinates, since $K_n=K^0_n\,\dot{\cup}\,K^1_n$, where $K^1_n$
consists of all tuples in $G^n$ with repeated coordinates. The
subrelation $K^0_n$ produces an algebra $\mathcal{K}^0_n$ with
$\displaystyle\frac{n!}{2}$ unary operations of even permutations
forming the alternating group $A_n$, and unary operations $r_b$
and $l_b$, for $b\in G$, carrying out the maps $a_i\mapsto a_ib$
and $a_i\mapsto ba_i$, respectively. Since all these operations
are invertible, with $(r_b)^{-1}=r_{b^{-1}}$ and
$(l_b)^{-1}=l_{b^{-1}}$, and form a {\em derivative} group
$S=S(G,n)$ with the identical even permutation which is equal both
to $r_e$ and $l_e$, we obtain an $S$-act $\mathcal{K}^0_n=\langle
K^0_n, s\rangle_{s\in S}$, for $|G|\geq n$. Here actions $r_b$ and
$l_b$ can be even permutations if, for instance, $|G|=n$, which
implies $S=A_n$, They may not belong to $A_n$, if $b\ne e$ and $G$
is torsion-free. And they coincide if $b$ belongs to the center of
$G$. The group $S$ is generated by its two subgroups $A_n$ and
$S'=\{r_b,l_b\mid b\in G\}$ and elements in $f\in A_n$ and in
$s\in S'$ commute: $fs=sf$. Besides, $r_b=l_b$ and
$\alpha_{bb'}=\alpha_b \alpha_{b'}$ for any $\alpha\in\{r,l\}$ and
$b,b'\in G$, if $G$ is commutative. Since $A_n$ is commutative iff
$n\leq 3$, the algebra $\mathcal{K}^0_n$ has the commutative
derivative group $S$ iff $n\leq 3$ and $G$ is commutative.

Clearly, if $G$ is a group $n$-spherically ordered by an order
$K_n$ then $G$ is $n$-spherically ordered by the dual order
$\overline{K_n}=(G^n\setminus K_n)\cup K^1_n$, too.
\end{remark}

In view of Theorem \ref{th_sp_cr} and Remark \ref{rem_sp_perm} we
have the following:

\begin{corollary}\label{cor_lin_cr}
A group $G$ is linearly ordered by a $2$-spherical order $K_2$ iff
for any pair $(a_1,a_2)\in K_2$ with $a_1\ne a_2$ and for any
$b\in G$ the pairs $(a_1b,a_2b)$ and $(ba_1,ba_2)$ are not
transpositions of some pairs in $K_2$.
\end{corollary}

Since all odd permutations for the group $S_3$ are transpositions
we ad\-di\-tion\-al\-ly have:

\begin{corollary}\label{cor_circ_cr}
A group $G$ is circularly ordered by a circular order $K_3$ iff
for any triple $(a_1,a_2,a_3)\in K^0_3$ and for any $b\in G$ the
triples $(a_1b,a_2b,a_3b)$ and $(ba_1,ba_2,ba_3)$ are not
transpositions of some triples in $K^0_3$.
\end{corollary}

\begin{remark}\label{rem_sp_perm2}\rm
Each orbit $O$ of the group $S$ on $K^0_n$ connects all even
permutations of tuples in $O$ and possibly tuples based on
distinct $n$-element sets if some $r_b$ or $l_b$ are not
permutations of these sets.

The number $r$ of these orbits $O$ is called the {\em rank of
generation} of $K^0_n$ with respect to $S$ and denoted by ${\rm
rk}_S(K^0_n)$. This rank is said to be the rank of generation for
$K_n$, denoted by ${\rm rk}_S(K_n)$.

If the group $G$ is abelian then  ${\rm rk}_S(K^0_n)$ is finite
iff $G$ is finite, since for tuples $(x_1,\ldots,x_n)$ and
$(y_1,\ldots,y_n)$ based on distinct $n$-element sets the link
$r_b=l_b$ is defined by $b=y_1-x_1$. For an infinite non-abelian
group one may put defining relations connecting $n$-tuples into
finitely many orbits via chains of tuples. Indeed, the connection
of tuples $(x_1,\ldots,x_n)$ and $(y_1,\ldots,y_n)$ can be
organized via inter\-me\-dia\-te tuples $(z_1,\ldots,z_n)$ formed
by new defining elements and with appropriate $r_b$ and $l_b$,
where the elements $b$ are composed with some new defining
elements. Thus there are infinite non-abelian groups with finite
${\rm rk}_S(K^0_n)$. Moreover, finite values ${\rm rk}_S(K^0_n)$
can be realized arbitrarily in $\omega\setminus\{0\}$ using
defining relations connecting step-by-step orbits by $r_b$ and
$l_b$ with new defining elements $b$.
\end{remark}

In view of Remark \ref{rem_sp_perm2} for an abelian group $G$ the
operators $r_b=l_b$ on $G^3$ either fix all coordinates of a
triple or move all its coordinates, i.e. can not generate odd
permutations (transpositions). Thus Corollary \ref{cor_circ_cr}
immediately implies:

\begin{corollary}\label{cor_circ_cr2}
For any abelian group $G$, $3\in{\rm Sp}_{\rm so}(G)$.
\end{corollary}

The following construction based on Theorem \ref{th_sp_cr} shows
how the spectrum ${\rm Sp}_{\rm so}(G)$ can be reduced till the
empty one, i.e. with $G$ which is not $s$-orderable in any way.

For a required group $G$ we take a generating set $\{a_n\mid
n\in\omega\setminus\{0\}\}\cup\{b,c\}$. Now we consider the
following defining relations: $ba_1c=a_2$, $ba_2c=a_1$,
$ba_nc=a_n$ for $n\geq 3$. These relations show that any $m$-tuple
$(ba_{i_1}c,ba_{i_2}c,\ldots,$ $ba_{i_m}c)$ consisting of pairwise
distinct elements and with some $i_j=1$ and $i_k=2$ is a
transposition of $(a_{i_1},a_{i_2},\ldots,a_{i_m})$. Therefore
$m$-spherical orders on $G$ coordinated with left and right group
actions can not be formed for any $m$. Hence we have the
following:

\begin{theorem}\label{th_sp_empty}
There exists a group $G$ such that ${\rm Sp}_{\rm
so}(G)=\emptyset$.
\end{theorem}

{\bf Definition.} Let $<$ be a strict linear order on a group $G$,
$C$ be the {\em cyclification} of $<$ consisting of all tuples
$(x,y,z)$ with $x<y<z\vee y<z<x\vee z<x<y$. A $n$-ary relation
$K^0_n$ on $G$, for $n\geq 3$, is called {\em $<$-coordinated} if
$K^0_n$ consists of all $n$-tuples $(a_0,\ldots,a_{n-1})$ with
$a_0<\ldots <a_{n-1}$, and for any $b\in G$ the tuples
$(a_0b,\ldots,a_{n-1}b)$ and $(ba_0,\ldots,ba_{n-1})$ satisfy
$C\left(a_ib,a_{(i+1)\,({\rm mod}\,n)}b,a_{(i+2)\,({\rm
mod}\,n)}b\right)$ and $C\left(ba_i,ba_{(i+1)\,({\rm
mod}\,n)},ba_{(i+2)\,({\rm mod}\,n)}\right),$ $i=0,\ldots,n-1$.

\medskip
\begin{remark}\label{ev_od}\rm By Remark \ref{rem_sp_perm} any $<$-coordinated relation $K^0_n$ is uniquely
ex\-ten\-si\-b\-le to its closure $K_n$ under even permutations of
tuples and addition of all tuples in $G^n$ with repeated
coordinates. In view of Remark \ref{rem_sp_perm} this closure
satisfies the axioms of $n$-spherical order iff even
per\-mu\-ta\-tions of tuples $(a_1,\ldots,a_n)$ in $K_n$ do not
meet their odd permutations under group actions
$(a'_1b,\ldots,a'_nb)$ and $(ba''_1,\ldots,ba''_n)$ in $K_n$,
where $(a'_1,\ldots,a'_n),(a''_1,\ldots,a''_n)\in K_n$, $b\in G$.
\end{remark}

\begin{remark}\label{rem_spher1}\rm
If $G$ is a finite group with $|G|=m$ and $m<n$ then $G$ is
$n$-spherically orderable. Indeed, in such a case the relation
$K_n$ consisting of all tuples with repeated coordinates satisfies
the axioms of $n$-spherical order implying that $K_n=K^1_n$ and
$\langle G,K_n\rangle$ is a $n$-spherically ordered group.
\end{remark}

In view of Remark \ref{rem_spher1} we have the following:

\begin{proposition}\label{pr_sp_fin}
If $G$ is a group with $|G|=m\in\omega$ then ${\rm Sp}_{\rm
so}(G)\supseteq\{n\in\omega\mid n>m\}$, in particular, $G$ is
almost totally $s$-orderable.
\end{proposition}

The following assertion is a reformulation of well-known folklore
fact that nonunit linearly ordered groups are infinite.

\begin{proposition}\label{pr_lo}
A finite group $G$ is $2$-spherically orderable iff $|G|=1$.
\end{proposition}

Proof. The one-element group $G$ is $2$-spherically orderable in
view of Remark \ref{rem_spher1}. If $|G|>1$ then $G$ contains a
nonunit element $g$ of finite order. Assuming that $G$ is
$2$-spherically orderable with the order $K_2$ we take the unit
$e$ and obtain $(e,g)\in K_2$ or $(g,e)\in K_2$. By the definition
of $2$-spherically ordered group we obtain $(g^k,g^{k+1})\in K_2$
for any $k$ or $(g^{k+1},g^{k})\in K_2$ for any $k$. For the first
case we take $k_0>0$ with $g^{k_0+1}=e$ and obtain $(g^{k_0},e)\in
K_2$, $g^{k_0}\ne e$, which contradicts the transitivity. In the
second case we also have a contradiction to the transitivity in
view of $(e,g^{k_0})\in K_2$, $g^{k_0}\ne e$. \endproof

\medskip
The arguments for Proposition \ref{pr_lo} imply the following:

\begin{corollary}\label{cor_lo1}
Any $2$-spherically orderable group $G$ is torsion-free, that is,
if $G$ contains an element of finite positive order then
$2\notin{\rm Sp}_{\rm so}(G)$.
\end{corollary}

\begin{remark}\label{rem_Z_m}\rm
Since each group $\mathbb Z_m$ is circularly orderable
\cite{Fuchs}, i.e. $3\in {\rm Sp}_{\rm so}(\mathbb Z_m)$, then
Proposition \ref{pr_sp_fin} and Co\-rol\-lary~\ref{cor_lo1} imply
that ${\rm Sp}_{\rm so}(\mathbb Z_2)={\rm Sp}_{\rm so}(\mathbb
Z_3)=\omega\setminus\{0,1,2\}$.

The groups $\mathbb Z_m$, for even $m$, are not $m$-spherically
orderable since the definition of $m$-spherically orderable group
implies that $K_m$ is closed both under even and odd permutations
of $m$ pairwise distinct elements: odd permutations are obtained
by actions of elements in $\mathbb Z_m$ on $m$-tuples of pairwise
distinct elements of $\mathbb Z_m$. It contradicts the axiom
(nso2). Thus $m\notin {\rm Sp}_{\rm so}(\mathbb Z_m)$ for each
even $m$. In particular, by Proposition \ref{pr_sp_fin} and
Co\-rol\-lary~\ref{cor_lo1}, ${\rm Sp}_{\rm so}(\mathbb
Z_4)=\omega\setminus\{0,1,2,4\}$.

The arguments above show that any group $G$ containing a subgroup
$\mathbb Z_m$, for even $m$, can not be $m$-spherically ordered by
a relation $K_m$ since, in view of (nso4), it should contain a
tuple $(i_1,\ldots,i_m)$, where
$\{i_1,\ldots,i_m\}=\{0,\ldots,m-1\}$. In particular, if $G$
contains subgroups $\mathbb Z_m$, for each even $m$, then ${\rm
Sp}_{\rm so}(G)$ is contained in the set of odd numbers.

At the same time the groups $\mathbb Z_m$, for odd $m\geq 5$, are
$m$-spherically orderable since the definition of $m$-spherically
orderable group implies even permutations of elements in $K_m$,
generated by the tuple $(0,1,\ldots,m-1)$ only. Thus $m\in {\rm
Sp}_{\rm so}(\mathbb Z_m)$ for each odd $m\geq 5$. \endproof
\end{remark}

In view of Proposition \ref{pr_sp_mon} and Remark \ref{rem_Z_m}
each element $a$ of even order $m$ in a given group $G$ implies
$m\notin{\rm Sp}_{\rm so}(G)$. Thus we have the following:

\begin{proposition}\label{pr_so_sp1}
For any group $G$, ${\rm Sp}_{\rm so}(G)$ does not contain even
numbers which are equal to orders of elements in $G$.
\end{proposition}

In particular, we have the following:

\begin{corollary}\label{pr_so_sp2}
If a group $G$ contains elements of each even order then $${\rm
Sp}_{\rm so}(G)\cap 2\mathbb Z=\emptyset.$$
\end{corollary}

\begin{proposition}\label{pr_so1}
For any natural $m, n$ with $2<n<m$, and $n\!\!\not|m$ if $n$ is
even, then the group $\mathbb Z_m$ is $n$-spherically orderable.
\end{proposition}

Proof. We form the $n$-spherical order $K_n$ on the universe
$\mathbb Z_m$ adding to the set of $n$-tuples with some repeated
coordinates all even permutations of tuples $(k_1({\rm
mod}\,m),\ldots,k_n({\rm mod}\,m))$ with $k_1<\ldots<k_n$ and even
per\-mu\-ta\-tions of tuples $(k_1+q({\rm
mod}\,m),\ldots,k_n+q({\rm mod}\,m))$ with $q\in\mathbb Z_m$, i.e.
$K_n$ is generated by its $<$-coordinated subrelation $K^0_n$,
where $<$ is the natural strict order on $\mathbb Z_m$.

Since $n\!\!\not|m$ for even $n$, $\mathbb Z_m$ does not contain a
subgroup $\mathbb Z_n$ violating the $n$-spherical orderability as
in Remark \ref{rem_Z_m}. Thus even permutations of tuples
$(k_1+q({\rm mod}\,m),\ldots,k_n+q({\rm mod}\,m))$ are coordinated
with even permutations of tuples $(k_1,\ldots,k_n)$, i.e. they do
not produce odd per\-mu\-ta\-tions. Since these per\-mu\-ta\-tions
cover all possibilities for tuples with even permutations only,
Theorem \ref{th_sp_cr} guarantees that $\mathbb Z_m$ is
$n$-spherically orderable by $K_n$.
\endproof

\medskip
Propositions \ref{pr_sp_fin}, \ref{pr_lo}, \ref{pr_so1} and Remark
\ref{rem_Z_m} immediately imply the following de\-scri\-p\-tion of
spherical spectra for the groups $\mathbb Z_m$:

\begin{theorem}\label{th_Z_m}
Let $m\in\omega\setminus\{0,1\}$. Then $${\rm Sp}_{\rm so}(\mathbb
Z_m)=\omega\setminus(\{0,1,2\}\cup\{n\mid n|m\mbox{ and }n\mbox{
is even}\}).$$
\end{theorem}

\begin{proposition}\label{pr_so1_tot}
The group $\mathbb Z$ is totally $s$-orderable.
\end{proposition}

Proof. Let $n\in\omega\setminus\{0,1\}$. We form the $n$-spherical
order $K_n$ on the universe $\mathbb Z$ adding to the set of
$n$-tuples with some repeated coordinates all even permutations of
tuples $(k_1,\ldots,k_n)$ with $k_1<\ldots<k_n$, i.e. generate
$K_n$ by the $<$-coordinated relation $K^0_n$ with the natural
order $<$. Clearly, for any $m\in\mathbb Z$,
$(k_1+m,\ldots,k_n+m)$ preserves the set of these tuples and
satisfies the axioms of $n$-spherical orders in view of Theorem
\ref{th_sp_cr}, as required. Thus, ${\rm Sp}_{\rm so}(\mathbb
Z)=\omega\setminus\{0,1\}$, i.e. $\mathbb Z$ is totally
$s$-orderable. \endproof

\medskip
It is known that any torsion-free abelian group can be
lexicographically ordered with respect to its generators.
Therefore the arguments for Pro\-po\-si\-tion \ref{pr_so1_tot}
imply the following:

\begin{theorem}\label{th_tf_tot}
Any torsion-free abelian group is totally $s$-orderable.
\end{theorem}

\begin{remark}\label{rem_tf}\rm
The group confirming Theorem \ref{th_sp_empty} is torsion-free and
it is not $s$-orderable at all. It illustrates that the
commutativity of a group is essential in Theorem \ref{th_tf_tot}.
\end{remark}





\end{document}
